Author's response to reviews

Title: Effectiveness of Motivational Interviewing in improving lipid level in patients with dyslipidemia assisted by general practitioners: Dislip-EM study protocol

Authors:

Luis Angel Pérula de Torres (luisangel.perula@gmail.com)
Josep Maria Bosch Fontcuberta (boschfontcuberta@gmail.com)
Julia Bóveda Fontán (jboveda1@gmail.com)
Manuel Campiñez Navarro (manuelcn@camfic.org)
Nieves Barragán Brun (apmotivacional@gmail.com)
Juan Carlos Arboniès Ortiz (juancarlos.arboniesortiz@osakidetza.net)
Jose Antonio Prados Castillejo (jaaprados@semfyc.es)
Enrique Martín Rioboó (enrique.martin@ono.com)
Remedios Martín Alvarez (rmartin@perevirgili.catsalut.net)
Josep Massons Cirera (jmassons.bnm.ics@gencat.net)
Margarita Criado Larumbe (margarita.criado.exts@juntadeandalucia.es)
Roger Ruiz Moral (roger.ruiz.sspa@juntadeandalucia.es)
Jose Angel Fernández Garcia (joseangelfernandezgarcia@hotmail.com)
Francisco Buitrago Ramirez (fbuitragor@meditex.es)
Inmaculada Olaya Caro (inmaculada.olaya.sspa@juntadeandalucia.es)
Modesto Pérez Díaz (osejamola@gmail.com)
Joaquin Ruiz de Castroviejo (jcastrov@terra.es)

Version: 8 Date: 2 November 2011

Author's response to reviews:

Dear sir,

Following your suggestions about the Manuscript entitled "Effectiveness of Motivational Interviewing in improving lipid level in patients with dyslipidemia assisted by general practitioners: Dislip-EM study protocol." (MS ID: 1577928128580913), we have made a point-by-point description of the changes included:

-Major revisions:

ü Author list: the list of authors in the manuscript is written now exactly as they appear in the submission system, both in style and order ("First name Initial Last name").

ü Affiliations: Department, Institute, (Street), City, (Postal code), Country have been included for every author.

ü Abstract: both abstract (the manuscript and the submission one) are now exactly the same.
Table: has been corrected.
Figures: have been removed.
Additional file has been removed from the manuscript.

-Minor revisions:
References: reference list has been placed after the competing interest section.
Corresponding author: is now included on the title page.
Figures: have been corrected as directed.
Typography has been revised.

Looking forward to hearing from you

Sincerely yours,

Luis A Pérula de Torres, MD
Unidad Docente de Medicina de Familia y Comunitaria de Córdoba
Servicio Andaluz de Salud
C/ Blanco Soler, 4
14004 Córdoba (Spain)